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Device Service Record Form for AC114
	Date
	Comments

(Include the following in each MSR entry: What, Where, Why)
	Info Provided By

	11/02/11
	Removed from crate in IB4 and transported to QA in IB4 in a music crate, gave cavity to Rob Riley 
	dburk

	Nov/04/2011
	Measured cavity on CMM.  Returned to crate.  
	R. Riley

	11-4-11
	Moved to ICB to optical inspection GS performed and completed inspection
	dburk

	11-4-11
	Moved to IB4 furnace area for TG testing
	dburk

	11/04/2011
	Large grain 9-cell cavity ACC114 moved to IB4 RF Lab.
	T. Khabiboulline

	11/07/2011
	F=1297.09 MHz, FF=99%, L=1057 mm. Cavity moved out of RF Lab.
	T. Khabiboulline

	11/07/11
	Cavity moved to CPL Tumbling Area
	G.Steuer

	12/13/11
	Prepping cavity for tumbling with 1st media ACT 3/8 x 3/8
	dburk

	12/13/11
	Tumbled for 4 hours removed media and used a modified LPR station with UPW and air pump to boost water pressure 
	Dburk

	12/14/11
	Prepping cavity for tumbling with RG 221/2 NEW cones
	Dburk

	12/15/11
	Started tumbling with the cones tumbled 11 hours today
	Dburk

	12/16/11
	Finished tumbling with 2 more hours for the total of 13 hours used UPW to clean out the cavity 
	Dburk

	12/19/11
	Prepped cavity for step 3 with wood blocks and 800 mesh al. ox. Started tumbling at 1320 ran for 7.5 hours
	Dburk

	12/20/11
	Continued  tumbling at 0655  
	Dburk

	12/21/11
	Stopped tumbling after getting 30 hours total on cavity cleaned with UPW and sent the cavity to be HPR
	Dburk

	12/22/11
	Moved cavity to CLP cleanroom and H.P.R.’d and left to dry in CPL cleanroom
	G.Steuer

	12/27/11
	Optical inspection needs to be done but the camera was moved and needs to have an ORC done before using
	Dburk

	1/3/12
	Moved cavity to IB4 tumbling area to prep for tumbling with step 4.
	G.Steuer

	1/4/12
	Began tumbling cavity with Step 4 (wood blocks and colloidal silica) 9am
	G.Steuer

	1/5/12
	Tumbling cavity with Step 4 (wood blocks and colloidal silica) continued 
	Dburk

	1/6/12
	Tumbling cavity with Step 4 (wood blocks and colloidal silica) continued
	Dburk

	1/9/12
	Stopped tumbling after 120hours of tumbling 
	Dburk

	1/9/12
	Complete Step 4 and low pressure rinsed the cavity then moved it to the CPL cleanroom and was H.P.R’d. It is now drying on the Backtech cart in the CPL cleanroom.
	G.Steuer

	1/9/12
	Moved cavity to RF lab where Timer will warm RF test this cavity tomorrow.
	G.Steuer

	01/10/2012
	F=1296.274 MHz, FF=94%, Ecc=0.27mm, L=1057.2mm.
	T. Khabiboulline

	1/11/12
	Moved cavity to ICB laser lab for Optical inspection.
	G.Steuer

	1/11/12
	Completed Optical inspection and moved cavity to ICB storage racks.
	G.Steuer

	1/13/12
	Cavity moved to CPL tumbling area for tour
	N. Bremer

	1/13/12
	Setup cavity on bench for tour
	Dburk

	1/13/12
	Placed cavity into crate after tour was finished
	Dburk

	1/27/12
	Moved cavity to ICB storage rack
	G.Steuer

	2/1/12
	Transported to ANL for 20um EP and HPR only
	Damon Bice

	2/7/2012
	Cavity received 20 µm EP at ANL
	R. Murphy

	2/8/12
	Cavity transported from ANL to IB4
	L. Zimmerman

	2/13/2012
	Cavity AC114 moved to IB4 RF Lab for RF QC. F=1296.42 MHz, FF=89.9%.
	T. Khabiboulline

	2/13/2012
	AC114 tuning done. F=129736MHz, FF=98.4%, Ecc=0.34mm.
	T. Khabiboulline

	2/14/2012
	Cavity transported from IB4 to ANL
	N. Bremer

	2/15/2012
	Cavity received 10 µm EP at ANL
	R. Murphy

	2/22/12
	Cavity received HPR / Assy sequence for VTS
	Damon Bice

	22Feb2012
	Cavity transported from ANL to IB1 for VTS
	N. Bremer

	03/06/12
	VTS test. Copy-and-paste of e-log entry by Dmitri:

The quench is in the cell #1. 

The max Q_0 (at 1.55K) was 4.6e10 (not that great). 

That was measured by second sound and verified by mode measurements. 

(Those are "raw" values of E_quench, not scaled with the field distribution) 

pi mode: E = 29 MV/m, Cell #1 

8/9 mode (max field in the cells 1 and 9): 32.1 MV/m, Cell 1 

2/9 mode (cells 3,7): 12.9 MV/m, cell 7 

3/9 mode (cells 2, 8): 15.5 MV/m, cell 8 

4/9 mode (cells 4, 6): 23.8 MV/m, cell 6
	Alex Melnychuk

	3/12/12
	Cavity moved to ICB for optical inspection.
	G.Steuer

	3/12/12
	Completed optical inspection. Moved cavity to ICB storage racks
	G.Steuer

	8/21/12
	Cavity location verification.  Cavity @ ICB.
	Jim Rife

	4/10/13
	Cavity location verification.  Cavity at ICB.
	Jim Rife

	1/14/14
	Cavity location verification.  Cavity location unknown.
	Jim Rife
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